Author's response to reviews

Title: Gene and cytokine profile analysis of macrolide-resistant Mycoplasma pneumoniae infection in Fukuoka, Japan

Authors:

Kentaro Matsuda (kentaromatsuda@hotmail.com)
Mitsuo Narita (naritamy@d5.dion.ne.jp)
Nobuyuki Sera (sera@fihes.pref.fukuoka.jp)
Eriko Maeda (e-maeda@fihes.pref.fukuoka.jp)
Hideaki Yoshitomi (yoshitomi@fihes.pref.fukuoka.jp)
Hitomi Ohya (ohya.dwhv@pref.kanagawa.jp)
Yuko Araki (araki_yuuko@med.kurume-u.ac.jp)
Tatsuyuki Kakuma (tkakuma@med.kurume-u.ac.jp)
Atsushi Fukuoh (fukuoh@junshin-u.ac.jp)
Kenji Matsumoto (matsumoto-k@ncchd.go.jp)

Version: 3
Date: 29 November 2013

Author's response to reviews: see over
Matsuda Children’s Clinic / National Center for Child Health and Development, Department of Allergy & Immunology

1-7-18 Shimoori Onojyo-city, Fukuoka 816-0952, Japan
Tel.: 092-501-2894, FAX: 092-582-9962, Email: kentaromatsuda@hotmail.com

November 29th 2013

**** COVER LETTER ****

BMC Infectious Diseases
Dear Philippa Harris

We are enclosing the manuscript entitled “Gene and cytokine profile analysis of macrolide-resistant Mycoplasma pneumoniae infection in Fukuoka, Japan” by Matsuda Kentaro, Narita Mitsuo, Sera Nobuyuki, Maeda Eriko, Yoshitomi Hideaki, Ohya Hitomi, Kakuma Tatsuyuki, Araki Yuko, Fukuoh Atsushi, and Matsumoto Kenji for consideration as a research article in BMC Infectious Diseases.

This work describes the present status of M. pneumoniae infection in Japan based on several laboratory findings. We believe the paper may interest your readers in particular because we observed a small outbreak of an M. pneumoniae genotype with a rarely-reported A2063T mutation and clarified its clinical features.

We warrant that all authors have seen and approved this version of the manuscript, and the article is original, is not under consideration by another journal, and has not been previously published.

Any correspondence should be directed to me at the following address:
Kentaro Matsuda, M.D., Ph.D.
Matsuda Children’s Clinic
1-7-18 Shimoori Onojyo-city, Fukuoka 816-0952, Japan
Tel.: +81-92-501-2894
FAX: +81-92-582-9962
Email: kentaromatsuda@hotmail.com

Thank you for your attention to our paper.

Sincerely,

Kentaro Matsuda, M.D., Ph.D.